

Contemporary Nurse



ISSN: (Print) (Online) Journal homepage: https://www.tandfonline.com/loi/rcnj20

Factors correlating with self-care behaviors among patients with coronary artery disease: a cross-sectional study

Imad Abu Khader, Malakeh Z. Malak, Mohammad Asia, Mohammed Jallad & Hisham Zahran

To cite this article: Imad Abu Khader, Malakeh Z. Malak, Mohammad Asia, Mohammed Jallad & Hisham Zahran (2023): Factors correlating with self-care behaviors among patients with coronary artery disease: a cross-sectional study, Contemporary Nurse, DOI: 10.1080/10376178.2023.2220424

To link to this article: https://doi.org/10.1080/10376178.2023.2220424





Factors correlating with self-care behaviors among patients with coronary artery disease: a cross-sectional study

Imad Abu Khader^a, Malakeh Z. Malak ^{b*}, Mohammad Asia^a, Mohammed Jallad^a and Hisham Zahran^a

^aFaculty of Nursing, Arab American University of Palestine (AAUP), Jenin, Palestine; ^bFaculty of Nursing, Al-Zavtoonah University of Jordan, Amman, Jordan

(Received 7 November 2022; accepted 26 May 2023)

Background: In Palestine, there is a lack of studies that examined self-care behaviors among patients with coronary artery disease in outpatient clinics.

Objective: This study purposed to evaluate self-care behaviors, examine the relationship between self-care behaviors and selected sociodemographic and psychosocial factors (e.g. depression, anxiety, stress, and social support), and self-efficacy, and determine predictors of self-care behaviors among patients with coronary artery disease in outpatient clinics in the West Bank/Palestine.

Design: A cross-sectional study was conducted.

Methods: A total of 430 Palestinian adult patients suffering from coronary artery disease attended outpatient clinics were recruited. A self-reported questionnaire consisting of the following tools: Depression, Anxiety, Stress Scale 21, Sullivian's Self-efficacy scale, and Multidimensional Social Support Scale was used to collect data during the period from the beginning of April to the beginning of July 2022. Descriptive and inferential statistics (Pearson's and Point-biserial correlation tests and multiple linear regression) were used for analyzing data.

Results: The patients reported low self-care behaviors levels and high self-efficacy levels. The psychosocial reactions endorsed by the patients were 86.3% for depression, 76.3% for anxiety, 43.3% for stress, and 98.6% had moderate and normal social support. A positive correlation was found between self-care behaviors and age (r = 0.160, p < 0.01), duration of disease (r = 0.095, p < 0.05), self-efficacy (r = 0.443, p < 0.01), and social support (r = 0.266, p < 0.01). Self-efficacy (B = 0.401, p < 0.01), social support (B = 0.160, p < 0.01), and age (B = 0.109, p < 0.05) were significant predictors of self-care behaviors in those patients.

Conclusion: Low self-care behaviors were a significant issue among patients with coronary artery disease in outpatient clinics. This study may help healthcare professionals develop health promotion programs for patients with coronary artery disease to improve self-care behaviors.

Keywords: anxiety; depression; self-care behaviors; self-efficacy; stress; social support

Impact statement

Healthcare professionals should develop health promotion programs that enhance self-care behaviors such as health education, self-care monitoring, maintaining a healthy diet, practicing

^{*}Corresponding author. Email: malakeh.m@zuj.edu.jo, malakehmalak@yahoo.com

regular physical activity, and adherence to prescribed treatment. Such programs should be directed to younger and newly diagnosed patients and consider social support and self-efficacy.

Plain language summary

This study aimed to assess self-care behaviors among adult patients with coronary artery disease and factors associated with it, in addition to significant predictors. There are many published studies about this problem among patients with cardiovascular diseases especially heart failure and hypertension, while there is a lack of studies in Palestine investigating this issue among adults with coronary artery disease. The current study could help healthcare professionals plan strategies to improve self-care behaviors among this cohort group. Four hundred and thirty adult patients with coronary artery disease in outpatient cardiac clinics responded to this study. A cross-sectional design was adopted and a convenience sample method was used to recruit participants. Data were collected using self-reported questionnaires that were distributed to the participants in the clinics. These findings can be discussed with healthcare professionals to enhance their awareness of this problem and help them suggest interventions to improve self-care behaviors among those patients.

Introduction

Cardiovascular diseases (CVDs) are one of the most serious health problems on a global level. The World Health Organization (WHO) reported 17.9 million annual deaths due to CVDs worldwide, which formed 32% of the total deaths (World Health Organization [WHO], 2021). Coronary artery disease (CAD) is one of these CVDs and is considered a major cause of mortality worldwide (Malakar et al., 2019). CAD refers to the lowering of blood supply to the heart due to plaque (cholesterol deposits and other elements in the arteries), which causes narrowing in the heart arteries and produces a partial or total blockage of the blood flow (Malakar et al., 2019). In Palestine, overall deaths among adults and older people are related to CVDs, especially Ischemic Heart Diseases (32.4% of all deaths) (Global Health Metrics, 2020).

Self-care refers to a decision-making process in selecting the behaviors of prevention and management of chronic illnesses and it includes three main elements: self-care maintenance, management, and monitoring (Riegel et al., 2017). Healthy behaviors are significant factors in reducing potential risks or complications of any illness and in assisting any effective treatments (Budreviciute et al., 2020). Self-care behaviors (SCBs) are due to therapeutic behaviors that should be performed by patients with CVDs. These behaviors include selecting a healthy diet, managing weight, taking prescribed medication, maintaining physician's appointments, demonstrating the recommended exercise, and monitoring symptoms (Tawalbeh et al., 2018). The CAD requires complex management involving specific medication and healthy SCBs.

Many studies have evaluated the SCBs among patients with CVDs and revealed different results, for example, Nemer (2022) found that Jordanian patients with CAD reported high levels of SCBs. While Asadi et al. (2019) revealed that Iranian patients with heart failure (HF) had a moderate level of SCBs. Peyman and colleagues (2018) found that Iranian patients with HF reported low SCBs level. Also, Tawalbeh et al. (2018) demonstrated that the levels of SCBs among Jordanian patients with HF were low.

Many factors are correlated with low levels of SCBs, including sociodemographic characteristics such as younger age (Ma, 2018; Mazar et al., 2020; Nemer, 2022; Peyman et al., 2018; Tawalbeh et al., 2018; Xie et al., 2020), gender as being male (Asadi et al., 2019), low educational levels (Asadi et al., 2019; Lee & Park, 2017; Tawalbeh et al., 2018), low income (Tawalbeh et al., 2018), and marital status as being single (Asadi et al., 2019; Nemer, 2022), as well as

short duration of the disease (Nemer, 2022; Tawalbeh et al., 2018). Additionally, psychological problems such as anxiety, stress, and depression could be negatively correlated with self-care behaviors (Bahari et al., 2019; Celano et al., 2016; Chen et al., 2020; Kim et al., 2020; Lee & Park, 2017; Nemer, 2022). Additionally, social support and self-efficacy are protective factors for adopting healthy SCBs (Bahari et al., 2019; Chen et al., 2020; Kim et al., 2020; Lee & Park, 2017; Ma, 2018; Nemer, 2022; Peyman et al., 2020).

Despite many published studies have been conducted on medication adherence among patients with chronic diseases (Al-Qerem et al., 2021; Al-Ramahi, 2015; Alsolami et al., 2015; Awad et al., 2017; Basheti et al., 2016) and among patients with CVDs, especially hypertension, and HF (Al-daken & Eshah, 2017; Alhaddad et al., 2016; Malak et al., 2021), few studies have been examined SCBs among patients with CAD at Arab countries scale (Nemer, 2022).

In Palestine, despite increasing the number of patients with CVDs, there is a lack of health-care services provided for this target group such as self-care counseling and monitoring, in addition to the unavailability of cardiac rehabilitation programs in Palestinian healthcare settings like outpatient clinics (Al-Aqtam et al., 2023). Furthermore, there has been a scarcity of research on the relationship between SCBs and patients with CAD. To address this gap, a national study was conducted to assess the health-promoting behaviors (HPBs) among patients with CAD in primary healthcare clinics. The results showed that over half of the participants either reported moderate (50.6%) or excellent (2.8%) HPBs. The overall lifestyle score of the participants was 123.7 (SD = 21.8), indicating a moderate lifestyle level, with the highest score in the nutrition subscale (M = 24.3, SD = 4.8) and the lowest score in the physical activity subscale (M = 15.2, SD = 4.4).

Thus, our study is considered one of the first studies in the West Bank/Palestine that discussed this problem and can help healthcare professionals (HCPs) develop strategies to enhance SCBs among patients with CAD. Therefore, this current study aimed to evaluate SCBs including levels, correlating factors, and predictors (e.g. sociodemographic, self-efficacy, and psychosocial (depression, anxiety, stress, and social support)) among patients with CAD in the West Bank/Palestine. Furthermore, these questions guided the study:

- (1) What are the prevalence levels of SCBs, self-efficacy, and psychosocial factors (depression, anxiety, stress, and social support) among adult patients with CAD in Palestine?
- (2) What is the association between selected sociodemographic variables (e.g. age, gender, educational level, marital status, income/month, and duration of disease) and the variables of self-efficacy, and psychosocial factors (such as depression, anxiety, stress, and social support) in adult patients with CAD in Palestine?
- (3) What are the significant predictors of SCBs among adult patients with CAD in the West Bank/Palestine?

Design, setting, population, and study sample

A cross-sectional, descriptive correlational design was used to carry out this study. This study was conducted at cardiac outpatient clinics located in West Bank Palestinian hospitals in two health sectors involving public and private hospitals in West Bank. In 2021, there were 15 public and nine private hospitals containing cardiac outpatient clinics (Palestinian Ministry of Health [MOH], 2021). The public clinics run under the responsibility of the Ministry of Health and private clinics run under the responsibility of each private hospital. These clinics provide follow-up services and treatment for patients with CVDs, where the patients have

4 I.A. Khader et al.

monthly appointments for monitoring their physiological measures and carrying out medical tests, in addition to obtaining prescribed medications. All patients with CAD should be followed up in these clinics.

The target population of this study was all adult patients with CAD attending the cardiac outpatient clinics in the hospitals. A cluster and stratified sampling methods were used to select nine outpatient cardiac clinics (6 public and three private), whereas the West Bank was divided into three zones: north, south, and middle. Then, a simple random method was used to select two public and one private outpatient cardiac clinic from each zone. A convenience sampling method was adopted to recruit the study sample. The G*power 3.1 software program (Heinrich Heine University, Dusseldorf, Germany) was utilized to calculate sample size with an alpha of 0.05, a small effect size of 0.05, and a power of 0.90 with ten predictors. According to regression, a total sample of 420 participants was needed to perform this study; the sample was increased to 462 to avoid incomplete questionnaires.

The eligible participants involved adult patients who were: (1) 20 years to less than 60 years old (because the physiological and psychological aspects and lifestyle patterns differ between adults and older adults according to the aging process), (2) experiencing CAD for at least six months, (3) having the ability to read and write, (4) not suffering from severe mental or cognitive disorders according to their medical file records to avoid the likelihood of worse CVD outcomes in these groups, and (5) willing to participate in this study. However, those patients who were admitted as inpatients were excluded from participation.

Study measurements

A structured self-reporting questionnaire was utilized to conduct this study. This questionnaire consists of the following measurements in addition to sociodemographic data (age, gender, educational level, marital status, monthly income, and duration of disease).

Self-care of Coronary Heart Disease Inventory (SC-CHDI) Version 3 was developed by Riegel and Dickson (2020) and consists of 23 items divided into three sub-scales, including self-care maintenance (9 items), management (8 items), and confidence (6 items). These items were rated on a five-point Likert scale ranging from 1 (never/rarely/not likely) to 5 (always/daily/ likely). The mean score for the total scale was used for determining an individual's level of self-care, with scores lower than the mean reflecting low self-care and scores equal to or greater than the mean indicating high self-care. This scale is valid and reliable, the original version of the scale, developed by Riegel and Dickson (2020), was found to have excellent internal reliability, with a Cronbach's alpha coefficient of 0.91. A subsequent study by Nemer (2022) adapted the scale to the Arabic language and also found it to be a valid and reliable measure, with a Cronbach's alpha of 0.824. These findings demonstrated the robustness of the self-care assessment tools chosen and their ability to accurately measure an individual's level of self-care.

Sullivan's cardiac self-efficacy scale (SCSES) was created by Sullivan et al. (1998) and composed of 13 items. These items were rated from 0 (not at all confident) to 4 (completely confident). This scale had two domains: control symptoms (eight questions) and maintenance functioning (five questions). The median for the total scale was used for identifying the patient's level of self-efficacy, with scores lower than the median (<26) indicating low self-efficacy and scores equal to or greater than the median (≥26) indicating high self-efficacy (Barham et al., 2019). This scale is valid and reliable, the original version of the scale, developed by Sullivan et al. (1998) demonstrated excellent internal reliability, with a Cronbach's alpha for control symptoms and maintenance functioning domains being 0.90 and 0.87, respectively. A later study by Barham et al. (2019) adapted the scale to the

Arabic language and also demonstrated it to be a valid and reliable measure, with a Cronbach's alpha of 0.848.

Depression Anxiety Stress Scales-21 (DASS-21). This scale was developed by Lovibond and Lovibond (1995) and revised by Antony et al. (1998) to assess psychological factors. It includes three subscales due to depression, anxiety, and stress, and each subscale consists of seven items. These items were rated on a 4-point Likert scale ranging from 0 (did not apply to me at all) to 3 (applied to me very much). The total score for each subscale ranged from 0 to 21, where higher scores indicated more likely to experience depression, anxiety, or stress. Earlier studies adapted the scale to the Arabic version and revealed its validity and reliability, with a Cronbach's alpha coefficient of 0.89 (Moussa et al., 2017).

Multidimensional Social Support Scale (MSPSS) was developed by Zimet et al. (1988) and consists of 12 statements to measure social support. The mean score for the total scale was used for determining the level of social support, where higher scores indicated high social support. This tool is valid and reliable, the original version of the scale, developed by Zimet et al. (1988) demonstrated excellent internal reliability, with a Cronbach's alpha of 0.88. A subsequent study by Alzayyat et al. (2015) adapted the scale to the Arabic language and demonstrated it to be a valid and reliable measure, with a Cronbach's alpha of 0.88.

Ethical considerations

The approval to perform this study was obtained from Institutional Review Board (IRB) at Arab American University at Palestine with reference No# 008/2022. Written informed consent, which was attached to each questionnaire, was obtained from eligible participants who were assured of confidentiality. It was made clear that participation was voluntary, where the participants can withdraw at any time during the study, and there would be no direct benefit or reward due to their participation.

Method of data collection

After obtaining approvals, the main researcher discussed the study's objectives with the head nurses of the cardiac outpatient clinics to facilitate data collection. Two researchers took the responsibility of distributing the questionnaires to eligible participants. The questionnaires were distributed to the patients attending these clinics and were collected at the same time during the period from the beginning of April to the beginning of July 2022. To ensure a high response rate, recurrent visits were arranged to the selected clinics on different days.

Data analysis

The "SPSS" statistical package system 25.0 was used to analyze the collected data. Data were assessed for normality, missing, and outliers, the results revealed that data were within normal and no outliers. Descriptive statistics including, mean, SD, and range were used to describe continuous variables (i.e. age, monthly income, duration of disease, SCBs, self-efficacy, depression, anxiety, stress, and social support), and frequencies and percentages were used to describe categorical variables (e.g. gender, educational level, and marital status). Pearson's correlation test was used to examine the relationship between continuous variables and the point-biserial test was used to assess the relationship between categorical and continuous variables. Multivariable linear regression was utilized to determine the predictors

(sociodemographic, psychosocial, and self-efficacy) of SCBs, in which the factors correlated with SCBs were entered into the regression analysis model. The level of significance was set at ≤ 0.05 .

Results

A total of 462 patients who registered in outpatient cardiac clinics were invited to participate in this study. However, 440 questionnaires were returned with a response rate of 95.2%. Additionally, ten questionnaires were dropped out for missing data or incomplete questionnaires, thus, the final number of questionnaires that were entered for analysis was 430.

Table 1 showed that the participants' mean age was M = 48.38 (SD = 7.9) with a range of 21–59 years. The majority of the participants (66.2%) were males, 90.9% were married, and 50.9% completed secondary education and less. The mean monthly income was 750.5 \$ (SD = 290.43) with a range of 282–2539\$. The duration of the disease mean was 4.0 years (SD = 4.61) with a range of 1–25 years.

Table 2 explained that the mean score of SCBs was 2.93 (SD = 0.47) and the median of self-efficacy was 37 [IQR $_{25-75}$ = 30.75–41.0]. These findings indicated that the participants reported low SCBs and high self-efficacy levels. Additionally, 43.3% of the participants reported experiencing stress, with 18.8% reporting severe to extremely severe stress. Furthermore, 76.3% of participants endorsed anxiety, with 46.2% experiencing severe to extremely severe anxiety. Additionally, 86.3% of participants reported experiencing depressive symptoms, with 47.3% reporting severe to extremely severe depressive symptoms. Also, 98.6% of participants reported having moderate to normal social support.

The results from Table 3 showed a significant positive relationship between SCBs and age (r = 0.160, p < 0.01), duration of disease (r = 0.095, p < 0.05), self-efficacy (r = 0.443, p < 0.01), and social support (r = 0.266, p < 0.01).

Concerning predictors of SCBs, the full model, which included four predictors of SCBs (age, duration of disease, self-efficacy, and social support), was statistically significant ($F_{(4.425)}$ = 33.438, p < 0.001, R = 0.489, $R^2 = 0.239$, adjusted $R^2 = 0.232$). This means that the model explained 23.9% of the variance in SCBs. Self-efficacy (B = 0.401, p < 0.01), social support

Characteristic	n	%	
Gender			
Male	284	66.0	
Female	146	34.0	
Educational level			
Secondary or less	219	50.9	
Higher than secondary	60	14.0	
Bachelor or higher	151	35.1	
Marital status			
Single	23	5.3	
Married	391	90.9	
Others (Divorce & widow)	16	3.7	
	M	SD	Range
Age	48.38 years	7.9	21–59 years
Income/month (\$)	750.5	290.43	282.0-2539.0
Duration of the disease	4.0 years	4.61	1–25 years

Table 1. Sociodemographic characteristics of the study sample (N=430).

Note: n: number; %: percentage; M: Mean; SD: Standard Deviation.

Table 2. Levels of self-care behaviors, self-efficacy, and psychosocial factors among study participants (N = 430).

Variables	$N\left(\%\right)$	M(SD)
Self-care behaviors		2.93 (0.47)
Sullivan's cardiac self-efficacy		,
Median = 37 [IQR ₂₅₋₇₅ = 30.75 -41.0]		
Stress	7.39 (5.38)	
Normal (0–7)	244 (56.7)	,
Mild (8–9)	39 (9.1)	
Moderate (10–12)	66 (15.3)	
Severe (13–16)	52 (12.1)	
Extremely severe (17 and above)	29 (6.7)	
Anxiety	,	7.66 (4.93)
Normal (0–3)	102 (23.7)	
Mild (4–5)	77 (17.9)	
Moderate (6–7)	52 (12.1)	
Severe (8–9)	41 (9.5)	
Extremely severe (10 and above)	158 (36.7)	
Depression	(, , ,)	10.10 (4.76)
Normal (0–4)	59 (13.7)	(,
Mild (5–6)	53 (12.3)	
Moderate (7–10)	115 (26.7)	
Severe (11–13)	97 (22.6)	
Extremely severe (14 and above)	106 (24.7)	
Social support	5.63 (0.93)	
M=1-2.9 (Low social support)	6 (1.4)	2.32 (0.50)
M = 3-5 (Moderate social support)	98 (22.8)	
M = 5.1-7 (High social support)	326 (75.8)	

Note: n: number; %: percentage; M: mean; SD: standard deviation. IQR: Interquartile range.

Table 3. Factors correlating self-care behaviors among study participants (N = 430).

Variables	Self-care behaviors		
	\overline{r}	<i>p</i> -value	
Age	0.160	0.001**	
Income/month	-0.037	0.447	
Duration of disease	0.095	0.040*	
Self-efficacy	0.443	0.000**	
Stress	-0.054	0.260	
Anxiety	-0.076	0.118	
Depression	-0.049	0.309	
Social support	0.266	0.000**	
	p.b.r	<i>p</i> -value	
Gender	-0.012	0.797	
Educational level	0.047	0.335	
Marital status	0.068	0.157	

Note: p.b.r: point-biserial correlation.

^{*}Correlation is significant at the \leq 0.05 level (2-tailed).

^{**}Correlation is significant at the ≤0.01 level (2-tailed).

(B = 0.160, p < 0.01), and age (B = 0.109, p < 0.05) were found to be significant predictors of SCBs in patients with CAD in outpatient clinics (as indicated in Table 4).

Discussion

This study purposed to assess SCBs levels, examine the factors correlating with SCBs, and predictors of SCBs among Palestinian adult patients with CAD in outpatient clinics. Our findings demonstrated that the participants had low SCBs, which is congruent with previous studies (Peyman et al., 2018; Peyman et al., 2020; Tawalbeh et al., 2018). However, it is inconsistent with a previous Jordanian study indicating higher levels of SCBs among adult patients with CAD (Nemer, 2022). This study's result might be related to a lack of healthcare services such as self-care counseling provided by HCPs and the unavailability of cardiovascular health promotion programs directed to improve SCBs among those patients in Palestinian outpatient clinics (Al-Aqtam et al., 2023). Financial issues could play a significant role in engaging in these behaviors (Cheng et al., 2015).

Our study found the majority of the participants endorsed high levels of depression, which is consistent with a previous Palestinian study (Allabadi et al., 2019). On the contrary, this study reported higher levels of depression among patients with CVDs compared with previous studies (Al-Abbudi et al., 2018; Bahall, 2019; Dhital et al., 2018; Ivanovs et al., 2018; Mbakwem et al., 2016; Umer et al., 2019; Yazew et al., 2019). This difference in results might be related to the characteristics of the study participants, most of our participants' mean age was around 48 years and completed secondary education or less. Besides, the majority of our participants reported a low income. These findings could have influenced their coping mechanisms with their diseases and highlighted the high levels of depression among those patients.

Our study showed that more than two-thirds of participants experienced anxiety. Our study's findings are incongruent with earlier studies indicating lower anxiety levels among patients with CVDs (Chen et al., 2019; Dhital et al., 2018; Ivanovs et al., 2018). The high anxiety levels among study participants might be associated with the determinants linked with CVDs including illnesses characteristics (Dhital et al., 2018), physiological alterations, unanticipated complications (Davies & Allgulander, 2013), required lifestyle changes (Celano et al., 2016), insufficient counseling resources (Dhital et al., 2018), and a lack of financial support (Dhital et al., 2018).

Also, more than one-third of the patients experienced stress, this percentage could be connected with high anxiety levels because both stress and anxiety affect one another. Thus, our findings promote implementing psychological management strategies for patients with CAD. Additionally, most of the participants endorsed moderate and normal social support, this expected study finding may be related to the collectivist culture in Palestinian society, where social support is critical and could play a role in enhancing SCBs.

95.0% CI Predictor h В t-test p-value Lower Age 0.006 0.109 2.467 0.014 0.001 Duration of disease 0.002 0.019 0.426 0.671 -0.007

3.657

9.208

0.000

0.000

0.160

0.401

Upper

0.012

0.011

0.125

0.028

0.038

0.018

Table 4. Predictors of self-care behaviors: Multivariable linear regression.

Note: b: Unstandardized beta; B: Standardized beta; CI: Confidence Interval.

0.081

0.023

Social support

Self-efficacy

Our finding revealed that participants reported high self-efficacy, which is higher than in previous studies among patients with CVDs (Barham et al., 2019; Ejadi et al., 2018; Lee & Park, 2017; Peyman et al., 2018). This high self-efficacy could be associated with many factors, including higher education (Khairy et al., 2021; Peyman et al., 2018), high social support (Chair et al., 2015), and the long duration of disease (Kang & Yang, 2013).

This study demonstrated a positive correlation existed between SCBs and age, indicating patients with advanced age reported higher SCBs. This study finding agreed with Ma's (2018) study, which found that middle-aged patients reported greater SCBs than young adults. Our result could be clarified as the majority of study participants were middle-aged adults who can make decisions and take responsibility for adopting SCBs (Strough & Bruine de Bruin, 2020). Also, age was considered a protective factor for SCBs in this study.

Our study showed that SCBs and duration of disease had a significant positive correlation, which suggested that patients with a longer duration of disease endorsed high SCBs. This result is consistent with previous studies (Tawalbeh et al., 2017, 2018). Our finding could be interpreted as patients experiencing CAD for a long time may have more experience with the disease and treatment, which could lead to greater adoption of SCBs (Beker et al., 2014).

Our study demonstrated a positive relationship between SCBs and self-efficacy, which illustrated as high self-efficacy motivated patients to enhance SCBs. This finding is consistent with previous literature (Chen et al., 2020; Lee & Park, 2017; Ma, 2018; Peyman et al., 2020). Self-efficacy is a drive for encouraging and motivating the adoption of healthy SCBs among patients with CAD (Wantiyah et al., 2020).

This study found a positive relationship between SCBs and social support, indicating that patients with high social support had better SCBs, which is congruent with earlier studies (Bahari et al., 2019; Lee & Park, 2017). Patients with high social support were more likely to seek advice from HCPs about SCBs (Gallagher et al., 2011). Additionally, family and friends can make a significant difference in a patient's health orientation, which may result in higher SCBs (Hu et al., 2015). Our study showed that social support was a predictor for SCBs, which agreed with a previous study (Tawalbeh et al., 2017). The potential interpretation of this finding might be related to societal norms in Arab society, where families can assist their patients during their therapeutic management (Padela & Zaidi, 2018).

Although this study is one of the first studies assessed SCBs among patients with CAD in Palestine, it has the following limitations; a cross-sectional design was adopted, where the variables were only measured over a short time; the responses relied heavily on the participants' perceptions at the time of data collection, therefore, other designs such as longitudinal could be used in the future studies.

Furthermore, the data were collected via self-reported questionnaires, which may have influenced the participants' desirability and resulted in response bias. Despite this study involving both health sectors and different geographical areas in Palestine, however, a convenience sampling method was adopted, which may affect the generalization of the findings. Also, the current study excluded older adults, illiterate people, and those with severe mental disorders, thus future studies are recommended to include these groups.

Impact of the study

This study can provide baseline information about SCBs and correlating factors among patients with CAD in Palestine, which may be significant for HCPs in setting priorities for planning healthcare programs to enhance SCBs in this target group. This study encourages HCPs to develop health promotion programs such as maintaining a healthy diet, practicing routine physical activity, adhering to prescribed medications, keeping scheduled appointments with

healthcare professionals, and self-care monitoring for those patients to improve SCBs. Earlier studies explained that health intervention programs provided by healthcare professionals in clinics enhanced SCBs (Peyman et al., 2020; Tawalbeh et al., 2017).

Conclusion

Patients with CAD in Palestine reported low levels of SCBs. There was a significant relationship between SCBs and psychosocial factors such as depression, anxiety, stress, social support, and self-efficacy. Thus, conducting screening for these psychological factors and implementing psychological management strategies for patients with CAD could enhance SCBs. HCPs could also leverage the social supports that are prevalent in Palestine's collectivist culture to enhance SCBs among Palestinian patients with CAD. Moreover, HCPs should develop and implement health promotion and education training programs focused on evidence-based practice to improve SCBs among patients with CAD. Future qualitative studies describing the SCBs' experiences, facilitators, and barriers to adhering to the SCBs of those patients are recommended. Moreover, studies investigating the effects of such management strategies and health promotion programs among this cohort group should be conducted.

Consent to participate

Each participant provided informed consent before beginning the study.

Funding

This research received no specific grant from any funding agency in the public, commercial, or not-for-profit sectors.

Disclosure statement

No potential conflict of interest was reported by the author(s).

Ethics approval

All procedures performed in studies involving human participants were in accordance with the ethical standards of the institutional and/or national research committee (Arab American University at Palestine) with reference No# 008/2022.

ORCID

Malakeh Z. Malak http://orcid.org/0000-0002-2412-2999

References

Al-Abbudi, S. J., Lami, F. H., & Wady, Z. A. (2018). Prevalence and assessment of severity of depression among ischemic heart disease patients attending outpatient cardiology department Baghdad Teaching Hospital, Baghdad, Iraq. *Journal of Psychiatry*, 21, 438. https://doi.org/10.4172/2378-5756.1000438

Al-Aqtam, I., Darawad, M., Alshraideh, J., Nabolsi, M., Shoqirat, N., & Ayed, A. (2023). Health-promoting behaviors among patients with coronary artery disease in Palestine. *Palestinian Medical and Pharmaceutical Journal (PMPJ)*, 8, 00–00. https://doi.org/10.59049/2790-0231.1142

- Al-daken, L., & Eshah, N. F. (2017). Self-reported adherence to therapeutic regimens among patients with hypertension. *Clinical and Experimental Hypertension*, 39(3), 264–270. https://doi.org/10.1080/10641963.2016.1247164
- Alhaddad, I. A., Hamoui, O., Hammoudeh, A., & Mallat, S. (2016). Treatment adherence and quality of life in patients on antihypertensive medications in a Middle Eastern population: Adherence. *Vascular Health Risk Management*, 12, 407–413. https://doi.org/10.2147/VHRM.S105921
- Allabadi, H., Probst-Hensch, N., Alkaiyat, A., Haj-Yahia, S., Schindler, C., Kwiatkowski, M., & Zemp, E. (2019). Mediators of gender effects on depression among cardiovascular disease patients in Palestine. *BMC Psychiatry*, 19(1), 284. https://doi.org/10.1186/s12888-019-2267-4
- Al-Qerem, W., Jarab, A. S., Badinjki, M., Hyassat, D., & Qarqaz, R. (2021). Exploring variables associated with medication non-adherence in patients with type 2 diabetes mellitus. *PLoS ONE*, *16*(8), e0256666. https://doi.org/10.1371/journal.pone.0256666
- Al-Ramahi, R. (2015). Adherence to medications and associated factors: A cross-sectional study among Palestinian hypertensive patients. *Journal of Epidemiology and Global Health*, 5(2), 125–132. https://doi.org/10.1016/j.jegh.2014.05.005
- Alsolami, F., Correa-Velez, I., & Hou, X. Y. (2015). Factors affecting antihypertensive medications adherence among hypertensive patients in Saudi Arabia. *American Journal of Medical Science*, *5*, 181–189. https://doi.org/10.5923/j.ajmms.20150504.07
- Alzayyat, A., Al-Gamal, E., & Ahmad, M. (2015). Psychosocial correlates of Internet addiction among Jordanian university students. *Journal of Psychosocial Nursing and Mental Health Services*, 53(4), 43–51. https://doi.org/10.3928/02793695-20150309-02
- Antony, M. M., Bieling, P. J., Cox, B. J., Enns, M. W., & Swinson, R. P. (1998). Psychometric properties of the 42-item and 21-item versions of the Depression Anxiety Stress Scales in clinical groups and a community sample. *Psychological Assessment*, 10(2), 176–181. https://doi.org/10.1037/1040-3590.10.2.
- Asadi, P., Ahmadi, S., Abdi, A., Shareef, O. H., Mohamadyari, T., & Miri, J. (2019). Relationship between self-care behaviors and quality of life in patients with heart failure. *Heliyon*, *5*(9), e02493. https://doi.org/10.1016/j.heliyon.2019.e02493
- Awad, A., Osman, N., & Altayib, S. (2017). Medication adherence among cardiac patients in Khartoum State, Sudan: A cross-sectional study. Cardiovascular Journal of Africa, 28(6), 350–355. https://doi. org/10.5830/CVJA-2017-016
- Bahall, M. (2019). Prevalence and associations of depression among patients with cardiac diseases in a public health Institute in Trinidad and Tobago. *BMC Psychiatry*, 19(1), 4. https://doi.org/10.1186/s12888-018-1977-3
- Bahari, G., Scafide, K., Krall, J., Mallinson, R. K., & Weinstein, A. A. (2019). Mediating role of self-efficacy in the relationship between family social support and hypertension self-care behaviours: A cross-sectional study of Saudi men with hypertension. *International Journal of Nursing Practice*, 25(6), e12785. https://doi.org/10.1111/ijn.12785
- Barham, A., Ibraheem, R., & Zyoud, S. H. (2019). Cardiac self-efficacy and quality of life in patients with coronary heart disease: A cross-sectional study from Palestine. *BMC Cardiovascular Disorders*, 19(1), 290. https://doi.org/10.1186/s12872-019-01281-7
- Basheti, I. A., Hait, S. S., Qunaibi, E. A., Aburuz, S., & Bulatova, N. (2016). Associations between patient factors and medication adherence: A Jordanian experience. *Pharmacy Practice*, 14(1), 639. https://doi. org/10.18549/PharmPract.2016.01.639
- Beker, J., Belachew, T., Mekonin, A., & Hailu, E. (2014). Predictors of adherence to self-care behaviour among patients with chronic heart failure attending Jimma University Specialized Hospital Chronic Follow up Clinic, South West Ethiopia. *Journal of Cardiovascular Diseases & Diagnosis*, 2, 180. https://doi.org/10.4172/2329-9517.1000180
- Budreviciute, A., Damiati, S., Sabir, D. K., Onder, K., Schuller-Goetzburg, P., Plakys, G., Katileviciute, A., Khoja, S., & Kodzius, R. (2020). Management and prevention strategies for non-communicable diseases (NCDs) and their risk factors. Frontiers in Public Health, 8, 574111. https://doi.org/10.3389/fpubh.2020.574111
- Celano, C. M., Daunis, D. J., Lokko, H. N., Campbell, K. A., & Huffman, J. C. (2016). Anxiety disorders and cardiovascular disease. *Current Psychiatry Reports*, 18(11). https://doi.org/10.1007/s11920-016-0739-5
- Chair, S. Y., Wong, K. B., Tang, J. Y.-M., Wang, Q., & Cheng, H. Y. (2015). Social support as a predictor of diet and exercise self-efficacy in patients with coronary artery disease. *Contemporary Nurse*, 51(2-3), 188–199. https://doi.org/10.1080/10376178.2016.1171726

- Chen, A. M. H., Yehle, K. S., Plake, K. S., Rathman, L. D., Heinle, J. W., Frase, R. T., Anderson, J. G., & Bentley, J. (2020). The role of health literacy, depression, disease knowledge, and self-efficacy in self-care among adults with heart failure: An updated model. *Heart & Lung*, 49(6), 702–708. https://doi.org/10.1016/j.hrtlng.2020.08.004
- Chen, Y. Y., Xu, P., Wang, Y., Song, T. J., Luo, N., & Zhao, L. J. (2019). Prevalence of and risk factors for anxiety after coronary heart disease: Systematic review and meta-analysis. *Medicine*, *98*(38), e16973. https://doi.org/10.1097/MD.0000000000016973
- Cheng, K., Ingram, N., Keenan, J., & Choudhury, R. P. (2015). Evidence of poor adherence to secondary prevention after acute coronary syndromes: Possible remedies through the application of new technologies. *Open Heart*, 2(1), e000166. https://doi.org/10.1136/openhrt-2014-000166
- Davies, S., & Allgulander, C. (2013). Anxiety and cardiovascular disease. *Mod Trends Pharmacopsychiatry*, 29, 85–97. https://doi.org/10.1159/000351945
- Dhital, P. S., Sharma, K., Poudel, P., & Dhital, P. R. (2018). Anxiety and depression among patients with coronary artery disease attending at a Cardiac Center, Kathmandu, Nepal. *Nursing Research and Practice*, 4181952, 1–6. https://doi.org/10.1159/000351945
- Ejadi, A., Rejeh, N., Heravi- Karimooi, M., & Tadrisi, S. (2018). Examining the level of self-efficacy and its related factors in patients with angina pectoris disease referring to selected hospitals in Tehran during 2018. *Journal of Critical Care Nurse*, 11, 9–16. https://doi.org/10.1155/2018/4181952
- Gallagher, R., Luttik, M. L., & Jaarsma, T. (2011). Social support and self-care in heart failure. *The Journal of Cardiovascular Nursing*, 26(6), 439–445. https://doi.org/10.1097/JCN.0b013e31820984e1
- Global Health Metrics. (2020). Global burden of 369 diseases and injuries in 204 countries and territories, 1990–2019: A systematic analysis for the Global Burden of Disease Study 2019. *The Lancet*, 369 (10258), 1204–1222. https://www.thelancet.com/journals/lancet/article/PIIS0140-6736(20)30925-9/fulltext#seccestitle190
- Hu, H. Li, L., & Arao, T. (2015). The association of family social support, depression, anxiety and self-efficacy with specific hypertension self-care behaviours in Chinese local community. *Journal of Human Hypertension*, 29(3), 198–203. https://doi.org/10.1038/jhh.2014.58
- Ivanovs, R., Kivite, A., Ziedonis, D., Mintale, I., Vrublevska, J., & Rancans, E. (2018). Association of depression and anxiety with cardiovascular co-morbidity in a primary care population in Latvia: A cross-sectional study. BMC Public Health, 18(1), 328. https://doi.org/10.1186/s12889-018-5238-7
- Kang, Y., & Yang, I. (2013). Cardiac self-efficacy and its predictors in patients with coronary artery diseases. *Journal of Clinical Nursing*, 22(17-18), 2465–2473. https://doi.org/10.1111/jocn.12142
- Khairy, S., Aslan, A., Samara, A. M., Mousa, I., Alkaiyat, A. S., & Zyoud, S. H. (2021). Factors associated with self-efficacy in patients with hypertension: A cross-sectional study from Palestine. *Journal of Health, Population and Nutrition*, 40(1), 1. https://doi.org/10.1186/s41043-021-00225-2
- Kim, A. S., Jang, M. H., Park, K. H., & Min, J. Y. (2020). Effects of self-efficacy, depression, and anger on health-promoting behaviors of Korean elderly women with hypertension. *International Journal of Environmental Research and Public Health*, 17(17), 6296. https://doi.org/10.3390/ijerph17176296
- Lee, E., & Park, E. (2017). Self-care behavior and related factors in older patients with uncontrolled hypertension. *Contemporary Nurse*, 53(6), 607–621. https://doi.org/10.1080/10376178.2017.1368401
- Lovibond, S. H., & Lovibond, P. F. (1995). *Manual for the depression anxiety stress scales*. Psychology Foundation of Australia.
- Ma, C. (2018). An investigation of factors influencing self-care behaviors in young and middle-aged adults with hypertension based on a health belief model. *Heart & Lung*, 47(2), 136–141. https://doi.org/10.1016/j.hrtlng.2017.12.001
- Malak, M., Al-thbetat, A., & AL-Amer, R. (2021). Psychosocial factors correlate with adherence to medications among cardiovascular outpatient clinics in Jordan. *Journal of Public Health*, 45(1), 206–213. https://doi.org/10.1093/pubmed/fdab356
- Malakar, A. K., Choudhury, D., Halder, B., Paul, P., Uddin, A., & Chakraborty, S. (2019). A review on coronary artery disease, its risk factors, and therapeutics. *Journal of Cellular Physiology*, 234(10), 16812–16823. https://doi.org/10.1002/jcp.28350
- Mazar, L., Salimabadi, Y., Nasirzadeh, M., & Safarian, E. (2020). Self-efficacy analysis of health promoting behaviors of hypertensive patients in Rafsanjan. *Pakistan Heart Journal*, 52. https://doi.org/10.47144/ phj.v52i4.1828
- Mbakwem, A., Aina, F., & Amadi, C. (2016). Depression in patients with heart failure: Is enough being done? *Cardiac Failure Review*, 2, 110. https://doi.org/10.15420/cfr.2016:21:1

- Moussa, M. T., Lovibond, P., Laube, R., & Megahead, H. A. (2017). Psychometric properties of an Arabic version of the Depression Anxiety Stress Scales (DASS). Research on Social Work Practice, 27(3), 375–386. https://doi.org/10.1177/1049731516662916
- Nemer, O. (2022). Self-efficacy and self-care behaviors among patients with coronary artery disease in Jordan. *International Research Journal of Pharmacy and Medical Sciences*, 5, 32–47.
- Padela, A. I., & Zaidi, D. (2018). The Islamic tradition and health inequities: A preliminary conceptual model based on a systematic literature review of Muslim health-care disparities. *Avicenna Journal of Medicine*, 8(1), 1–13. https://doi.org/10.4103/ajm.AJM 134 17
- Palestinian Ministry of Health (MOH). (2021). Health annual report, Palestine. http://moh.ps/Content/Books/NWNJXX7RJ92Bn4f5EGYiH43a2tjAAzKBnseGnEUCaqWYZndsbCcPy_JOWguvkHTR4Xk4zUpdT45ooWxH11BhIbVAxwpGWy2wiwHdGcM5K7aZ.pdf
- Peyman, N., Abdollahi, M., & Zadehahmad, Z. (2018). The study of related factors with self-care and self-efficacy in heart failure patients. *Journal of Torbat Heydariyeh University of Medical Sciences*, 6, 55–61
- Peyman, N., Shahedi, F., Abdollahi, M., Doosti, H., & Zadehahmad, Z. (2020). Impact of self-efficacy strategies education on self-care behaviors among heart failure patients. *Journal of Tehran Heart Center*, 15, 6–11.
- Riegel, B., & Dickson, V. V. (2020). *Self-care of coronary heart disease inventory—Patient version 3* (English). http://self-care-measures.com/available-self-care-measures/self-care-ofcoronary-heart-disease-inventory/
- Riegel, B., Moser, D. K., Buck, H. G., Dickson, V. V., Dunbar, S. B., Lee, C. S., Lennie, T. A., Lindenfeld, J., Mitchell, J. E., Treat-Jacobson, D. J., & Webber, D. E. (2017). Self-care for the prevention and management of cardiovascular disease and stroke. *Journal of the American Heart Association*, 6, e006997. https://doi.org/10.1161/JAHA.117.006997
- Strough, J., & Bruine de Bruin, W. (2020). Decision making across adulthood. *Annual Review of Developmental Psychology*, 2(1), 345–363. https://doi.org/10.1146/annurev-devpsych-051120-010038
- Sullivan, M. D., LaCroix, A. Z., Russo, J., & Katon, W. J. (1998). Self-efficacy and self-reported functional status in coronary heart disease. *Psychosomatic Medicine*, 60(4), 473–478. https://doi.org/10.1097/ 00006842-199807000-00014
- Tawalbeh, L., Al Qadire, M., Ahmad, M. M., Aloush, S., Abu Sumaqa, Y., & Halabi, M. (2017). Knowledge and self-care behaviors among patients with heart failure in Jordan. *Research in Nursing & Health*, 40, 350–359. https://doi.org/10.1002/nur.21805
- Tawalbeh, L., Al-Smadi, A., AlBashtawy, M., AlJezawi, M., Jarrah, M., Al-Mahasees, A., & Aloush, S. (2018). The most and the least performed self-care behaviors among patients with heart failure in Jordan. Clinical Nursing Research, 29, 108–116. https://doi.org/10.1177/1054773818779492
- Umer, H., Negash, A., Birkie, M., & Belete, A. (2019). Determinates of depressive disorder among adult patients with cardiovascular disease at outpatient cardiac clinic Jimma University Teaching Hospital, South West Ethiopia: Cross-sectional study. *International Journal of Mental Health Systems*, 13(13). https://doi.org/10.1186/s13033-019-0269-8
- Wantiyah, W., Saputra, M. R., & Deviantony, F. (2020). Self-efficacy and health status in coronary artery disease patients. *Jurnal Ners dan Kebidanan Indonesia*, 15, 14. https://doi.org/10.20473/jn.v15i1. 17628
- World Health Organization. (2021). Cardiovascular diseases (CVDs). Retrieved June 11, 2021, from https://www.who.int/news-room/fact-sheets/detail/cardiovascular-diseases-(cvds)
- Xie, Z., Liu, K., Or, C., Chen, J., Yan, M., & Wang, H. (2020). An examination of the socio-demographic correlates of patient adherence to self-management behaviors and the mediating roles of health attitudes and self-efficacy among patients with coexisting type 2 diabetes and hypertension. *BMC Public Health*, 20(1), 1227. https://doi.org/10.1186/s12889-020-09274-4
- Yazew, K. G., Beshah, D. T., Salih, M. H., & Zeleke, T. (2019). Factors associated with depression among heart failure patients at cardiac follow-up clinics in Northwest Ethiopia, 2017: A cross-sectional study. Psychiatry Journal, 2019, 1–8. https://doi.org/10.1155/2019/6892623
- Zimet, G. D., Dahlem, N. W., Zimet, S. G., & Farley, G. K. (1988). The multidimensional scale of perceived social support. *Journal of Personality Assessment*, 52(1), 30–41. https://doi.org/10.1207/s15327752jpa5201 2